Author's response to reviews

Title: Retention of Healthcare Professionals in Urban India Using Health Information Technology - A Path Analysis Study

Authors:

Indrajit Bhattacharya (indrajitbhat@gmail.com)
Anandhi Ramachandran (anandhirama@gmail.com)

Version: 3
Date: 20 March 2014

Author's response to reviews:

Added:
1. Competing Interests
2. Author Contribution
3. Acknowledgements